Toward the Estimation of Oil Slick Thickness on Newly Formed Sea Ice Using C-Band Radar Backscatter

Elvis Asihene^(D), *Graduate Student Member, IEEE*, Alexander S. Komarov, *Senior Member, IEEE*, Gary Stern^(D), Colin Gilmore^(D), *Senior Member, IEEE*, and Dustin Isleifson^(D), *Senior Member, IEEE*

Abstract-Oil thickness in oil spills involving sea ice is a key parameter required for an effective oil spill response; however, quantifying it from radar backscatter data remains a difficult task. We investigated a possible solution for estimating oil slick thickness by using electromagnetic (EM) forward and inverse scattering models of oil-covered newly formed sea ice (NI). Our forward model employs a first-order approximation of a multilayered small perturbation method (SPM) to predict two copolarization C-band radar backscatters of NI covered by an oil slick with thicknesses ranging from 0 to 7 mm. The results showed that the backscatter decreases as slick thickness increases, which we attributed to signal attenuation within the saline-oil layer. Our inverse model relies on the particle swarm optimization (PSO) algorithm to determine the slick thickness on NI using synthetic backscatter data, and it requires the input of several important ice and oil physical parameters (thickness, dielectrics, and roughness). Moreover, the estimated slick thickness was validated using scatterometer data from an oil-on-ice experiment at the University of Manitoba's Sea-ice Environmental Research Facility (SERF). With synthetic data, the 5 mm oil slick thickness was overestimated by 25%, while with experimental data, it was overestimated by 8%. Overall, our findings have laid the groundwork for future inversion studies to identify the thickest oil spill zone from current and future C-band radar satellites for immediate response.

Index Terms—Arctic, backscatter estimation, inversion, modeling, oil slick, oil spill, oil thickness, particle swarm optimization (PSO), radar, sea ice, small perturbation method (SPM).

I. INTRODUCTION

THE risk of an oil spill in the Arctic has been increasing due to the unprecedented reduction of sea ice in human history, which has led to more shipping and resource extraction activities [1], [2], [3], [4]. To mitigate this risk, nearcontinuous monitoring is needed to timely detect and respond

Elvis Asihene and Gary Stern are with the Centre for Earth Observation Science (CEOS), University of Manitoba, Winnipeg, MB R3T 5V6, Canada (e-mail: asihenee@myumanitoba.ca).

Alexander S. Komarov is with the Data Assimilation and Satellite Meteorology Research Section, Environment and Climate Change Canada, Ottawa, ON K1A 0H3, Canada.

Colin Gilmore and Dustin Isleifson are with the Centre for Earth Observation Science (CEOS) and the Department of Electrical and Computer Engineering, University of Manitoba, Winnipeg, MB R3T 5V6, Canada. Digital Object Identifier 10.1109/TGRS.2024.3350891

to such hazards in this remote and difficult to access region. Satellite-based microwave remote sensing (e.g., synthetic aperture radar, SAR) is a well-established tool that shows strong potential for the task [5], [6]; however, the complex physical interaction between the sea ice and oil makes the microwave interpretation of oil spills extremely difficult. This is where electromagnetic (EM) modeling process comes into play and it has two parts [7]. The first part is forward modeling, where the material parameters are known and we seek the expected EM fields, and this allows us to understand how the physics of oil-ice interactions relate to the microwave scattering behavior. The second part is inverse modeling, where we take EM measurements and attempt to infer the material properties that caused those measurements.

Inversion could allow us to estimate oil slick thickness that may be distributed beneath, within, and on top of the ice, but it becomes particularly difficult because different combinations of physical properties of layered media (such as roughness and dielectrics) can cause similar backscatter signatures. This non-uniqueness is especially bad when considering scatterometer measurements, which can have limited scanning geometry and a single transmit/receive position (i.e., monostatic radar).

The ultimate goal of this manuscript is to estimate oil slick thickness on experimental oil-covered newly formed ice (NI) using ground truth C-band scatterometer data, which is particularly relevant for calibrating contemporary C-band radar satellites, including the Canadian RADARSAT Constellation Mission (RCM). To achieve this, we seek to answer two research questions as follows:

- 1) What is the expected C-band radar backscatter response when simulated oil slick thickness on NI increases?
- 2) How feasible is it to estimate the slick thickness on NI using the experimentally measured C-band radar backscatter?

The first question is addressed by employing a multilayered-SPM forward solver to simulate the total backscatter response of oil-covered nilas ice, while the second question is addressed by using an inverse solver (that calls the forward model) based on particle swarm optimization (PSO) algorithm. Synthetic backscatter data will be generated to evaluate the performance of the proposed PSO algorithm. As a validation step, we use scatterometer data from an oil-on-ice experiment at the University of Manitoba's Sea-ice Environmental Research Facility (SERF).

© 2024 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/

Manuscript received 23 August 2023; revised 4 December 2023; accepted 3 January 2024. Date of publication 8 January 2024; date of current version 23 January 2024. This work was supported in part by GENICE, in part by the Research Manitoba, in part by the Canada Research Chair (CRC) Programs, in part by the Natural Sciences and Engineering Research Council (NSERC) of Canada, and in part by the Canada Foundation for Innovation (CFI). (*Corresponding author: Elvis Asihene.*)

The remainder of this manuscript is organized as follows. Section II introduces key physical and radar aspects of oilcovered NI. Section III provides an overview of our EM modeling strategy used to estimate slick thickness on the ice surface. Section IV presents the results of the forward and inverse scattering models of oil-covered NI. Section V discusses the model sensitivity to changes in input oil parameters, along with the oil thickness retrieval. Finally, Section VI concludes with a summary and recommendations.

II. THEORETICAL BACKGROUND

To accurately determine the thickness of an oil slick on NI, it is important to first understand how microwave energy interacts with sea ice and oil as a function of their geophysical and thermodynamic properties. This section provides concise background information on oil-covered NI, including their dielectric mixture model, scattering measurement/model, and slick thickness retrieval.

A. Oil-Covered Newly Formed Sea Ice (NI)

NI is a complex multiphase material consisting mostly of ice crystals and brine inclusions. This type of sea ice is prevalent during freeze-up seasons and exists in various formations (such as frazil, grease ice, nilas, and pancake ice [8]), and its surfaces can be bare, covered with frost flowers, or covered with snow [9]. Under relatively calm ocean conditions, NI usually grows into two distinct ice types: dark nilas (<5 cm thick) and light nilas (5–10 cm thick) [10]. The topmost layer of these ice types is characterized by a highly saline frazil [11], making the NI profile a stratified material [12].

In the event of an oil spill on Arctic NI-covered waters, weathering processes are expected to occur at a slower rate or remain unchanged due to subzero ambient temperatures [13]. This causes the spilled oil to spread and drift slowly over the ice terrain, creating a much thicker slick layer suitable for effective response methods such as in situ burning and chemical dispersants.

B. Dielectric Mixture Model of Oil-Covered NI

Using dielectric mixture models, the dielectric properties corresponding to each layer of the NI profile could be derived from the ice geophysical properties. The refractive dielectric mixture model has consistently shown agreement with direct dielectric measurements among various models used to calculate sea ice dielectrics [14]. In the case of oil contamination in sea ice, the presence of oil introduces a new component within the ice structure, necessitating the inclusion of oil in the dielectric mixture model [15], [16]. If the spilled oil forms a slick cover solely on the NI, it creates a heterogeneous mixture of oil and brine due to the high salinity of the topmost NI layer [17]. The precise mixing process remains uncertain, as the hydrophobic nature of oil suggests that it will remain separate from the liquid brine, which is primarily water [18]. As such, it is appropriate to employ two distinct constituent materials in the refractive mixture model, a technique also applicable to mixtures in the liquid phase [19].

C. Scattering Measurement of Oil-Covered NI

Sea ice backscatter refers to the signal that is returned when microwave radiation interacts with the surface and medium of the ice. Its strength is influenced by three key factors [20]: the ice surface roughness, measured by the root-mean-square (rms) height and correlation length; the ice complex dielectric constant (CDC), which depends on its geophysical properties; and the microwave system parameters, such as frequency, incidence angle, and polarization. Meanwhile, the penetration depth of the EM wave is mostly governed by the ice dielectric constant and radar frequency [21]. To quantify the amount of backscatter per unit area of an ice scene, we utilize a metric called the Normalized Radar Cross Section (NRCS) [22]. Calculating the ratio of this metric in VV (vertical transmit-vertical receive) to HH (horizontal transmit-horizontal receive) polarization will reduce the surface roughness effect on the backscatter response [23].

NRCS data can be obtained through direct observation using remote microwave sensors, such as SAR and scatterometer. For oil-contaminated scenes, the measured scattering responses are very low and close to the sensor's noise floor, known as noise-equivalent sigma zero (NESZ) [24]. This leads to a low signal-to-noise ratio (SNR) of the NRCS values [25]. According to Minchew et al. [24], at least 6 dB above the NESZ level will enhance the SNR of the measured NRCS for oil-on-open water. Similar studies have proposed thresholds ranging from 7 to 9 dB [26], and even up to 10 dB [27]. However, in case of oil-on-ice, there is no detailed SNR analysis due to the lack of a major oil spill in ice-covered water regions. As a result, Johannson et al. [28] utilized two paired SAR images of oil-contaminated open water and oilfree NI, applying 2 dB threshold for the SNR analysis, while acknowledging the influence of noise on signal below the 6 dB threshold. We intend to use the same approach for our SNR analysis.

D. Scattering Model of Oil-Covered NI

Another way of obtaining NRCS data is via simulation, achieved by employing forward scattering models, which are classified into surface and volume components [29], [30]. Because of the high salinity of the topmost layer of NI [12], [31], radar signals usually interact with the ice surface, making surface scattering the dominant scattering mechanism.

Several mathematical models have been developed to predict ice-scattering mechanisms dominated by surface scattering. These include the small perturbation method (SPM), Kirchhoff approximation, integral equation method, and computational EM techniques [29], [32], [33], [34]. Of these, SPM is highly suitable for predicting the scattering of NI because it is computationally inexpensive and can simulate radar scattering of a slightly rough surface (that could be submerged in the medium), such that the vertical variation (rms height, h) and horizontal variation (correlation length, l) are both smaller than the incident wavelength (λ). This means that SPM is applicable under specific roughness conditions, where $h < 0.05\lambda$, $l < 0.5\lambda$, and h/l < 0.3 [35]. Even in situations where oil covers the ice and smoothens the surface, the SPM conditions still hold true.

Given the stratified nature of nilas ice, a multilayered SPM model will be useful in modeling the total scattering responses. Komarov et al. [36] developed a first-order approximation of the multilayered-SPM and used it to simulate VV and HH polarization backscatter of snow-covered sea ice systems when the dominant scattering occurs at the air/snow, snow/ice, and ice/seawater interfaces [36], [37], [38]. Isleifson et al. [15] successfully applied this model to various oil-contaminated sea ice systems, with oil distributed beneath the ice. In our case, we implement this multilayered-SPM in a scenario where oil is on top of the ice.

E. Retrieval of Slick Thickness on Oil-Covered NI

Turning to how oil slick thickness on NI is retrieved, inverse models rely on optimization algorithms to minimize the objective function between the modeled and measured NRCS data [39]. The process involves repeated calls to forward models during the iterative minimization. Prior to the actual inversion process, synthetic NRCS is often generated to test the effectiveness of such optimizers. Retrieving oil thickness on sea ice is extremely difficult because no operational radar for an oil spill incident in the Arctic is available. This limitation can be addressed by conducting experimental studies to collect in situ radar scattering of oil-contaminated sea ice. To date, only two studies by Firoozy et al. [40] and [41], have utilized a differential evolution optimizer to retrieve the oil layer thickness entrapped beneath and within ice. In contrast, our study employs a PSO algorithm to retrieve the oil layer thickness on top of the ice. Experimental studies in this research domain are scarce, and attempts to extract slick thickness when spilled oil spreads over sea ice scenes have not yet been fully explored, motivating our investigation.

III. MATERIAL AND METHODS

A. Input Datasets

Table I provides the input physical datasets used to drive the EM modeling strategy and they were collected along with C-band scatterometer data during the SERF 2020 oil-on-sea ice experiment [17]. Although the details of this experiment have been discussed elsewhere in [17], [42], and [43], it is worth noting that the sea ice was grown in a cylindrical tub of 4-m diameter and 1-m depth, under two phases of oil spill scenarios. In phase-1, oil was injected beneath existing ice, and in phase-2, the tub was filled with artificial seawater contaminated with 6.75 cm³ of Tundra Crude (a crude oil type provided by Tundra Oil & Gas in Canada). This article focuses on the phase-2 experiment. It is important to note that we did not actively control the thickness of the oil slick. Instead, the oil thickness naturally formed a skim layer over the established ice pack, with an average thickness of 5-mm measured from core samples.

During the experiment, we measured the in situ physical characteristics of ice and crude oil, which is required for calculating the CDCs of the spilled oil, oil-contaminated ice, and uncontaminated ice. These dielectric values, along Throughout the experiment, our C-band polarimetric microwave scatterometer was mounted on a 5.3 m scaffolding tower and scanned over the tub at a 24.5° incidence angle with respect to the nadir without interference from the edges of the tub. The measured backscatter values were VV = -39 dB, HH = -42 dB, and VH = -46 dB [17]. These values remained relatively constant above the NESZ level of -45 and -50 dB for the co- and cross-polarization signals, respectively. As a result, we expect a poor SNR for our measured NRCS that could affect the retrieval parameters, including the oil thickness and dielectric property [44]. It is worth noting that we used only the measured NRCS in VV and HH polarizations because our NRCS simulation model is limited to copolarization channels (see Section IV).

Our scatterometer has a very low NESZ compared to the current satellite SAR (e.g., RCM ScanSAR low noise beam mode [45]), which compelled us to use the 2 dB threshold from Johannson et al. [28] as a "mild threshold" above the NESZ for both modeled and measured NRCS data, ensuring their suitability for subsequent analysis. Despite this, we acknowledge the 6 dB threshold recommended by Minchew et al. [24] as a "harsh threshold" to avoid any possible noise contamination from the receiver noise background [24]. Details on our scatterometer system and calibration methods are provided in the literature [46].

B. Electromagnetic (EM) Modeling Strategy

Fig. 1 provides an overview of the EM modeling strategy used to determine the thickness of the oil slick on oil-covered NI (i.e., the target). The procedure begins with the inputs of parameters (see Table I) into our multilayered-SPM forward solver, which simulates C-band radar with VV and HH polarizations and 20° – 60° incidence angles. To serve the purposes of our forward solver, we vary a number of input parameters. In Fig. 1(a), we adjusted only the oil thickness from 1 to 7 mm to predict the expected NRCS of the target. In Fig. 1(b), we modified certain parameters as unknowns (e.g., oil thickness, oil dielectrics, and ice dielectrics) and fixed certain parameters as a-prior information (e.g., ice thickness, roughness, and water dielectrics) for each forward solver that our PSO-based inverse solver called. During the inversion process, the output is fed into the PSO algorithm which attempts to find the parameter of interest (oil layer thickness) by modifying the unknowns until the modeled NRCS matches the experimental NRCS.

Using synthetic NRCS ($\sigma^{o(\text{syn})}$ in dB units) data serves as a preliminary step to test the applicability of the PSO and to fine-tune its hyperparameters. This better prepares our inverse solver for validation with ground truth scatterometer NRCS measurements of experimental oil spills on artificially developed saline ice. During the testing of the inversion process with $\sigma^{o(\text{syn})}$, we added 1% Gaussian white noise (G_n) to the

OIL-FREE NI PROFILE	MEASURED VALUES	INPUT VALUES	OIL-COVERED NI PROFILE	MEASURED VALUES	INPUT VALUES
air	-	ε ₀ =1	air		ε0=1
air/ice interface			air/oil interface		
rms height (h)	0.18–0.44 cm	$h_{02} = 0.2 \ cm$	rms height		$h_{01}=0\ cm$
correlation length (l)	0.89–1.04 cm	$l_{02} = 1 \text{ cm}$	correlation length		$l_{01} = 0 \text{ cm}$
frazil			saline-oil		
thickness (đ)	0.5 cm	$d_2 = 0.5 \text{ cm}$	thickness	5 mm	$d_1 = 1 - 7 \text{ mm}$
temperature	-13°C		temperature	-13°C	
salinity	37.5 PSU		salinity	37.5 PSU	
pure ice CDC	3.177+j0.0004*	$\epsilon_2 = 6.67 + j2.64$	weathered oil CDC	2.27+j0.02 [†]	\succ $\epsilon_1 = 26.53 + j26.21$
brine volume	$18\%^{*}$		brine volume	18%*	
brine CDC	34.90+j36.94*		brine CDC	34.90+j36.94*	
sea ice			oil/ice interface		
thickness	4.5 cm	$d_3 = 4.5 \text{ cm}$	rms height		$h_{12} = 0.05 \text{ cm}$
temperature	-7°C		correlation length		$l_{12} = 1 \text{ cm}$
salinity	16 PSU		oiled-frazil		
pure sea ice CDC	3.182+j0.0005*	$\epsilon_3 = 5.82 + j1.76$	thickness	0.5 cm	$d_2 = 0.5 \text{ cm}$
brine volume	12%*		temperature	-13°C	
brine CDC	43.16+j43.22*		salinity	37.5 PSU	
ice/water interface			pure sea ice CDC	3.18+j0.0004*	$s_{0} = 5 14 \pm i1 62$
rms height		$h_{3w} = 0.2 \text{ cm}$	oil volume	30%	2 C ₂ 5.14+j1.02
correlation length		$l_{3w} = 1 \text{ cm}$	brine volume	18%*	
seawater			brine CDC	34.90+j36.94*	
temperature	−2°C	57 52 1140 00	sea ice		
salinity	32.8 PSU	$-\epsilon_{\rm w} = 5/.53 + 140.80$	ice/water interface	(The same with o	il-free ice)
			seawater		

 TABLE I

 Inputs Data for the Electromagnetic Modeling Strategy

The superscript, * means that the pure sea ice complex dielectric constant (CDC), brine CDC, and brine volume were calculated using formulas developed by Mätzler and Wegmüller [22], Cox and Weeks [47], and Stogryn and Desargent [22], respectively. Whereas, the superscript, \dagger implies that the CDC of the weathered oil was obtained from [41]. The subscripts, 1, 2, and 3 denote oil, frazil, and sea ice layers, respectively. For example, ε_1 represents the effective CDC of the saline-oil layer. Meanwhile the subscripts, 0 and w, represent the upper air and lower seawater half-spaces. The brace shapes show the parameters used for calculating the corresponding CDCs in the refractive mixture formula [22]. Note that we assumed the following values: 1) for the brine volume and brine CDC of the saline-oil layer due to the complex process of brine and oil mixtures [17]; and 2) for the roughness at the oil/ice (h₁₂, l₁₂) and ice/water (h_{3w}, l_{3w}) interfaces because there was no access for direct measurement.

modeled NRCS ($\sigma^{o(\text{mod})}$ in dB units) at 20°–60° incidence angles (θ):

$$\sigma_{pq}^{o(\text{syn})}(\theta) = \sigma_{pq}^{o(\text{mod})}(\theta) + G_n(r)N_f \tag{1}$$

$$N_f = \max_{\theta=20^{\circ}}^{60^{\circ}} \left| \frac{\sigma_{\rm VV}^{o(\rm mod)}(\theta) + \sigma_{\rm HH}^{o(\rm mod)}(\theta)}{2} \right| \qquad (2)$$

where the subscripts p and q denote the linearly transmitted and received polarizations, which are orthogonal to each other as vertical (V) or horizontal (H) components, respectively. The notation, r represents the independent normally distributed random numbers at VV and HH polarizations. Note that the noise was added to avoid overly optimistic estimates of inversion performance on synthetic data, and the exact magnitude of that noise was a choice made by matching the noise until the synthetic data resembled the experimental data. 1) Forward Solver: This study utilizes a forward solver based on the first-order approximation of multilayered SPM to simulate the backscatter of an oil-covered NI profile. This forward solver was developed by Komarov et al. [36] and has been widely used for applications that require robust prediction of multilayered media, such as snow-covered sea ice [36], [37], [38] and oil-contaminated sea ice [15]. Full details can be found in [36].

In the multilayered-SPM framework, the idealized ice profile is a multilayer medium between the upper air and lower water half-spaces. We set up this framework with two different ice profiles, as shown in Fig. 2. Each layer is characterized by specific parameters, such as thickness (d), CDC (ε), and roughness statistics (h and l). The first case pertains to oilfree NI, which necessitates the calculation of NRCS from two rough interfaces (air/ice and ice/water). This model case



Fig. 1. Overview of electromagnetic modeling strategy used for estimating oil thickness on newly formed sea ice: (a) forward solver and (b) inverse solver.



Fig. 2. Newly formed sea ice (NI) multilayer profile: (a) oil-free and (b) oil-covered. All the notations are defined in Table I.

provides a baseline for comparison with the oil-contaminated ice. In contrast, the second case involves an oil-covered NI, requiring the calculation of NRCS from two rough interfaces (oil/ice and ice/water). In this case, we varied the oil layer thickness on NI, in the range of 1–7 mm. The oil-covered NI has two significant aspects. First, we designate the slick oil on NI as "saline-oil" due to its contamination with brine that migrated from the highly saline topmost layer of NI. Second, we refer to the frazil layer as "oiled-frazil" because it has been contaminated by the spilled oil. Table I shows the input parameter values used in our multilayered-SPM forward solver. The pure sea ice CDC, brine CDC, and brine volume were calculated using formulas developed by Mätzler and Wegmüller [22], Cox and Weeks [47], and Stogryn and Desargent [22], respectively. Whereas, the CDC of the weathered oil was obtained from [41]. For each layer, the effective CDC was computed using the refractive mixture formula [22]. The upper half-space is air with real dielectric constant, $\varepsilon_0 = 1$, while the lower half-space is seawater with CDC, ε_w derived from Klein and Swift formulas [22].

In the monostatic setup, the cross-polarization channel within the first-order multilayered SPM is zero; therefore, the results presented in Section III-A include the copolarization channel in VV and HH polarizations. Using only two copolarization channels limits the polarization diversity of the NRCS. Such diversity is crucial for deriving polarimetric parameters used to discriminate scenes with low backscatter contrast or strong volume scattering mechanisms (e.g., see [48] and [49]). However, this limitation is not an issue in this study, which focuses on a bare oil-covered ice scene predominately influenced by surface scattering mechanisms. The cross-polarization backscatter in this case is very low and could be at the NESZ level, making it not usable for inversion. As a result, in an operational setting under our study, having copolarization backscatter measurements across multiple incidence angles is more important for inversion than having measurements from all polarizations at a single incidence angle.

To help interpret the effect of surface roughness on the radar scattering, we calculated the copolarization ratio, $\Upsilon_{VV/HH}$, which represents the ratio of NRCS in VV to HH polarization. Within the first-order approximation of SPM, in the case

of a single rough interface, this ratio is independent of the roughness parameters and only depends on the ice dielectrics and radar incidence angle [23]. In addition, to explore how far the microwave radiation propagates through the oil-covered NI profile, we computed the microwave penetration depth for each layer given by the formula in [21].

2) Inverse Solver: Once the scattering properties have been modeled, it serves as the basis for inverse solvers to estimate the unknown physical parameters from the observed scattering data. In this study, the inverse solver relies on the PSO algorithm (a global optimizer inspired by the social behavior of bird flocks, updating their position and velocity to achieve an optimum search solution [50]).

To estimate the oil slick thickness over the NI profile, the PSO algorithm seeks to minimize the objective function, which is a measure of the estimation error difference, ϵ_{σ^o} , based on the inputs between the experimental (or synthetic) and modeled backscatter data [see Fig. 1(b)]. Thus, the objective function, f, can be expressed in two different ways as (3) and (4), shown at the bottom of the page.

In (3), the objective function, f_1 , uses NRCS data for the minimization, where $\sigma^{o(*)}$ (in dB units) represents either the synthetic NRCS when testing the performance of PSO or the experimental NRCS when validating PSO. In (4), the objective function, f_2 , uses the ratio of NRCS in VV to HH polarization data for the minimization, where $\Upsilon_{VV/HH}^{(*)}$ (in dB units) follows a similar definition as the NRCS in (3). The symbol, τ represents three low incidence angles (θ) of 20°, 22°, and 25° when using synthetic inputs. Conversely, τ represents only one incidence angle, 24.5°, when using experimental inputs. The notations, $\vec{\chi}$ and $\vec{\chi}'$ represent vectors of the unknown and known variables of the oil-covered NI (see Table I for the definitions of these variables).

The unknown variables include d_1 , ε_1 , ε_2 , and ε_3 , which are the retrievable variables to be optimized by PSO for the oil-covered NI profile. This indicates that PSO must initialize seven unknowns for the minimization (note that each dielectric medium, ε is complex, with both real and imaginary parts). Meanwhile, the known variables include d_2 , d_3 , h_{01} , h_{12} , h_{3w} , l_{01} , l_{12} , and l_{3w} , are a priori information based on the delineation of certain ice types obtained from SAR satellite imagery. Incorporating these input datasets substantially reduces the number of unknowns and improves the estimation accuracy. Nevertheless, we acknowledge that they may not be entirely precise because, in reality, no operational SAR can directly extract the oil/ice interface roughness (e.g., h_{12} , l_{12} , h_{3w} , and l_{3w}) and underlying ice thickness (d_2 and d_3) which

TABLE II Input Parameter Values for the PSO Inverse Solver Implemented in MATLAB

Parameter		Value
Swarm size		70
Number of unknown variables		7
Weight inertia, w (linearly decreased)		1.1-0.1
Acceleration coefficients, c_1 and c_2		1.49 per each
Error tolerance for termination		10^{-6}
Hybrid function		"fmincon"
Maximum No. of function evaluations		1400
Unknown variables	Lower bound	Upper bound
$d_1^*(mm)$	0.5	10
ϵ_1 ; ϵ_1 "	22; 22	52; 42
ε_2 ; ε_2 "	3.5; 0.5	6; 2
ε ₃ '; ε ₃ "	3.5; 0.5	6; 2

The symbol * means that oil slick thickness, d_1 (mm), is the estimated variable of interest. The notations, ε_1 , ε_2 , and ε_3 , denote the effective CDCs for the oil layer, frazil layer, and sea ice layer, respectively (see Table I). The tick and double tick superscripts are the real and imaginary of CDC.

are obscured by the surface oil cover [17]. This compelled us to test the effect of incorrect ice roughness and thickness parameters on the estimation accuracy in our inversion process. For instance, we will adjust the experimentally measured value of d_3 from 4.5 cm to 2.5 and 8.5 cm, and check the estimation accuracy of the 5 mm oil thickness (see Table V in Section IV-B for the adjustments on other subsurface parameter). Our adjustments were constrained by the following: 1) the SPM region of validity: $h < 0.05\lambda$, $l < 0.5\lambda$, and h/l <0.3 [35] and 2) the thickness ranges of frazil and nilas, which are 0.1–1 cm and 2–10 cm, respectively [10].

In this inversion framework, the PSO algorithm was implemented in MATLAB [51], with parameterization values provided in Table II. Note that the lower and upper bound of these unknown variables were estimated from [21], [41], [52], [53]. We set the lower bound of oil thickness, d_1 slightly above zero to ensure that oil must be present before the algorithm works.

IV. RESULTS

Following the implementation of the EM modeling strategy described in Section III, we present our results in two

$$f_{1}(\vec{\chi}, \vec{\chi}') = \min(\epsilon_{\sigma^{o}}) = \sum_{\theta=\tau} \frac{\left|\sigma_{VV}^{o(*)}(\theta) - \sigma_{VV}^{o(mod)}(\vec{\chi}, \vec{\chi}'; \theta)\right|^{2} + \left|\sigma_{HH}^{o(*)}(\theta) - \sigma_{HH}^{o(mod)}(\vec{\chi}, \vec{\chi}'; \theta)\right|^{2}}{\left|\sigma_{VV}^{o(*)}(\theta)\right|^{2} + \left|\sigma_{HH}^{o(*)}(\theta)\right|^{2}}$$
(3)

$$f_2(\vec{\chi}, \vec{\chi}') = \min(\epsilon_{\sigma^o}) = \sum_{\theta=\tau} \frac{\left| \Upsilon_{\text{VV/HH}}^{(*)}(\theta) - \Upsilon_{\text{VV/HH}}^{(\text{mod})}(\vec{\chi}, \vec{\chi}'; \theta) \right|^2}{\left| \Upsilon_{\text{VV/HH}}^{(*)}(\theta) \right|^2}$$
(4)



Fig. 3. Modeled C-band backscatter from an oil-covered NI: (a) angular response of co-polarization NRCS at varying oil thickness, (b) thickness response of co-polarization NRCS at 25° and 50° incidence angles, and (c) angular response of co-polarization ratio. For comparison, we included the experimentally measured C-band VV and HH polarizations (VV^(meas) and HH^(meas)), collected from a 5-mm-thick layer of oil on top of the established NI [17].

subsections that correspond to our research questions posed in Section I. First, we delve into the NRCS results derived from the forward solver based on the multilayered SPM. Three sets of modeled backscatter datasets with varying slick thicknesses on ice will be shown. Second, we investigate the retrieval of oil slick thickness from the inverse solver based on our PSO-based parametric inversion process. We will showcase two inversion results: the near-ideal case based on synthetic data to evaluate the suitability of the algorithm and the real case based on experimental scatterometer measurement from the SERF 2020 oil-on-ice experiment.

A. Modeled Backscatter Results

Fig. 3 depicts the angular response of our forward solver, which predicted the C-band backscatter from an oil-covered NI at various oil slick thicknesses (ranging from 1 to 7 mm). Nonetheless, only for thicknesses of 1, 3, and 5 mm are the NRCS values well above the NESZ level (i.e., -45 dB) for incidence angles $<35^{\circ}$.

To avoid poor SNR, we set two thresholds: a mild threshold of 2 dB and a harsh threshold of 6 dB above the NESZ [see Fig. 3(a) and (b)]. Employing the mild threshold, the results showed that the predicted NRCS corresponding to 1, 3, and 5 mm slick thicknesses remained above the "NESZ + 2 dB" (i.e., -43 dB) for incidence angles $<30^{\circ}$. Using the harsh threshold, a significant portion of the predicted data exceeded the "NESZ + 6 dB" (i.e., -39 dB) for incidence angles $<25^{\circ}$. The only exception was a few points in the HH polarization for the 5-mm slick layer, which marginally fell below this threshold. Concerning the measured NRCS data that represents only the 5-mm oil-covered NI, both VV and HH exceeded the mild threshold. However, when the harsh threshold was imposed, the VV polarization met the limit, whereas the HH polarization fell 3 dB below it. In summary, our SNR analysis indicates that NRCS data at lower incidence angle ($\leq 25^{\circ}$) will be reliable for the subsequent retrieval analysis.

In Fig. 3(a), we observe a downward linear trend in the modeled NRCS of the oil-free NI (0 mm oil thickness) as the incidence angle increases from 20° to 60°. This behavior is expected, especially with the VV polarization showing a higher backscatter level and a gentler slope from -12 to -19 dB, in contrast to the HH polarization, which exhibits a lower backscatter level and a steeper slope from -13to -29 dB. These oil-free NRCS data, both in trend and magnitude, serve as baseline for comparing the NRCS data predicted from the oil-covered NI. When we simulated a 1 mm oil slick layer over the NI surface, the expected NRCS values for VV and HH polarizations decreased slightly by 9 and 8 dB, respectively. However, with 3 and 5 mm slick thicknesses, the predicted copolarization NRCS values experienced a significant drop, averaging 19 and 26 dB, respectively. To validate the accuracy of our forward solver prediction, we compare between the modeled and measured NRCS for oil slick thickness of 5 mm. The modeled VV completely captures the experimental measurement, while the corresponding HH compares well within 2 dB difference. In general, there was a clearly defined contrast once oil is introduced into the NI scattering simulations.

Fig. 3(b) shows the oil thickness response of the modeled NRCS at incidence angles of 25° and 50° . As expected, the NRCS exhibited a continuous decline with increasing oil thickness, and also the predicted VV and HH signatures were more closely related at 25° compared to the 50° incidence angle.

In Fig. 3(c), we show a different insight into the predicted backscatter response, which is the angular response of the ratio of modeled NRCS in VV to HH polarization (denoted as $\Upsilon_{VV/HH}$). The oil-free $\Upsilon_{VV/HH}$ values increase from ~ 1 to 10 dB, and upon introducing oil, these values inconsistently

TABLE III SIMULATED MICROWAVE PENETRATION DEPTH THROUGH EACH LAYER OF THE OIL-COVERED NI PROFILE

Layer	Penetration Depth (mm)
Saline-oil	2
Oiled-frazil	12
Sea ice	12

decrease by 2, 1.5, and 2 dB for slick layers measuring 1, 3, and 5 mm, respectively. Comparing this with the simulated NRCS data leads to two notable observations. First, both the oil-free and oil-covered $\Upsilon_{VV/HH}$ data exhibit an increasing trend as the incidence angle ranges from 20° to 60°. Second, as oil slick thickness increases, there is no discernible contrast.

Table III demonstrates the simulated penetration depth when microwave radiation propagates through each layer of the oil-covered NI profile. Based on the results, we expected two behaviors of microwave penetration. First, when the oil thickness is less than 2 mm, the microwave energy successfully penetrates through the saline-oil layer and oiled-frazil layer until it reaches the sea ice medium. Second, when the oil thickness is 2 mm or greater, the microwave penetration becomes limited within the saline-oil medium, leading to NRCS values closer to or below the NESZ level. Despite this, we noticed that the backscatter response at a 25° incidence angle remained above the NESZ level (including the mild and harsh thresholds) until the surface oil thickness approached 5 mm [see Fig. 3(b)].

B. Inversion Results

Table IV presents the results of the PSO estimation of oil slick thickness and the related CDCs in different layers of the oil-covered NI profile. Hereafter, the term "True" refers to simulated values, as the actual oil thickness (d_1) was simulated, except for $d_1 = 5$ mm, which we directly measured during our experimental work. Note that the elapsed run time took less than 5 min to complete.

With the synthetic NRCS data at three lower incidence angles $(20^\circ, 22^\circ, \text{ and } 25^\circ)$, we noticed a consistent pattern where the estimated oil thickness generally increased with the true oil thickness. For instance, when the true oil thickness was 1 mm, the estimated value closely approximated 0.84 mm. Likewise, for oil thicknesses of 3 and 5 mm, the estimated values proportionately increased to 2.62 and 6.23 mm, respectively. While the corresponding retrievable CDCs exhibited random variations as the oil thickness increased, they were held relatively close to their true values within a reasonable margin of error.

With the synthetic $\Upsilon_{VV/HH}$ at the same incidence angles, we noticed an unacceptable discrepancy between the true oil thickness and the estimated oil thickness. Moreover, the corresponding retrievable CDCs remained unchanged as the oil thickness increased, and their estimated values were held relatively close to their true values.



Fig. 4. Inverted oil slick thickness: (a) estimation error difference between the true and estimated oil thickness values, based on the synthetic (Syn) and experimental (Exp) NRCS data. The vertical line is a separation between "Syn" and "Exp" data; and (b) objective function evaluation.

With the experimentally measured NRCS data at a single incidence angle (24.5°) , we noticed that the estimated slick thickness closely approximated the true value. Similar to the synthetic NRCS, the accompanying retrievable CDCs ranged reasonably close to their true values.

Fig. 4 depicts the estimation error difference and the convergence of the objective function based on our primary estimated variable of interest (i.e., oil thickness). In Fig. 4(a), we compare the error difference between the true and estimated oil thickness values. For the synthetic NRCS, the error difference increases with thicker oil layers; specifically, the 1 and 3 mm oil thicknesses were underestimated by 3% and 8%, respectively, whereas the 5 mm oil thickness was overestimated by 25%. For the experimental NRCS, the error difference was found to be 8% overestimated based on the 5 mm oil thickness.

Fig. 4(b) demonstrates how the objective function progressively minimizes the difference between the modeled and experimental NRCS. With each iteration, the function evaluation consistently decreased, eventually reaching the optimal global minimum ($f_1 = 0.000432$) at iteration 62.

Table V presents the retrievable variables of the oil-covered NI profile using incorrect subsurface known ice thickness and roughness data. Our primary estimated variable of interest is the 5 mm oil thickness. For the oiled-frazil layer, we observed that the retrieved oil thickness closely matched the true value (within 0.7 mm average difference). For the sea ice layer, we found that the estimated oil thickness closely aligned with the actual value, with an average difference of 0.4 mm. Regarding the subsurface roughness, we tested two cases: 1) independently adjusting either the h (marked with the superscript) or l (marked with the ** superscript) and 2) simultaneously adjusting both h and l values (marked with the † superscript). At the oil/ice interface, when we make independent adjustments, the retrieved oil thickness closely matched the true value (within 0.5 mm average difference). However, with the simultaneous adjustments, the estimated

Data	Incidence angle	'True' oil thickness	Output of Inversion				
		đ ₁ (mm)	đ ₁ (mm)	ε ₁ ('True'=26.53+j26.21)	ε ₂ ('True'=5.13+j1.62)	ε ₃ ('True'=5.82+j1.76)	(secs)
Dual co-polarization NRCS (VV and HH)							
Synthetic		1	0.84	24.75+j35.55	4.58+j1.05	4.62+j0.62	191
Synthetic	20, 20, 25	3	2.61	29.42+j33	4.57+j0.94	4.77+j0.60	176
Synthetic		5	6.23	43.33+j25.37	4.82+j1.01	4.44+j0.71	252
Experimental	24.5	5*	5.38	52+j34.95	6+j2	3.50+j0.50	72
Co-polarization	n ratio, $\Upsilon_{VV/HH}$						
Synthetic		1	1.15	22+j22	6+j0.50	4.40+j0.50	125
Synthetic	20, 22, 25	3	1.61	52+j22	6+j0.50	4.67+j0.50	81
Synthetic		5	1.61	52+j22	6+j0.50	4.67+j0.50	88

TABLE IV ESTIMATED VARIABLES OF THE OIL-COVERED NI PROFILE

The term 'True' denotes simulated values, as the actual oil thickness (\mathfrak{d}_1) was simulated, except for $\mathfrak{d}_1 = 5$ mm, which we directly measured during our experimental work. The bold highlight means that the primary estimated variable of interest is \mathfrak{d}_1 . The notations, ε_1 , ε_2 , and ε_3 , denote the CDCs for the saline-oil layer, oiled-frazil layer, and sea ice layer, respectively (see Table I for more details). Note that the elapsed run time was based on standard PC specifications with 8 GB RAM and 1.6 GHz 4-core CPUs.

oil thicknesses were deemed unacceptable. On the contrary, at the ice/water interface, all the estimated oil thickness results closely aligned with the actual value, differing by an average of 0.1 mm, irrespective of the independent or simultaneous adjustments.

V. DISCUSSION

Our primary objective in this manuscript has been to describe how C-band radar data can be used to retrieve oil thickness on top of NI terrains. Two questions are raised to address this objective: what is the expected C-band radar backscatter when a simulated oil slick thickness on the NI surface increases? And how feasible is it to retrieve the slick thickness on NI using experimentally measured C-band radar backscatter? It is clear that the answers to these questions touch upon the issues of modeling backscatter sensitivity, which is a forward scattering problem, and oil spill retrieval, which is an inverse scattering problem.

A. Modeled Backscatter Sensitivity

To address the first question, it is important to distinguish between the backscatter characteristics of oil-free NI and oil-covered NI. Our study presents a novel approach of using a first-order multilayered-SPM forward solver to predict the scattering behavior of stratified bare, undeformed NI; nonetheless, remarkable progress has been achieved with other forward solvers ([12], [31], and references therein). For instance, Nghiem et al. [12] used a multilayer scattering model that considered both surface and volume contributions to predict the C-band scattering response of the 8 cm bare NI profile, following similar stratification shown in Fig. 2(b). Their modeled results predicted copolarization NRCS, which decreases over the range of incidence angles with VV having a gentler slope and a higher value than HH. In comparison, our multilayer scattering predictions, which only account for surface contributions, agreed with the decreasing trend of their NRCS, except that our NRCS magnitude was 10 dB higher. This discrepancy is due to the differences in experimental setups, ambient environmental conditions and scattering models.

In fact, it is reasonable to assert that modeling the C-band backscatter of bare oil-free NI is relatively well-established, and any remaining unclear scattering mechanisms can be readily identified and are likely to be addressed in the near future. However, when the NI surface is covered with frost flowers, snow, or a combination of both, considerably more work remains to be done, particularly modeling of frost flowercovered NI, which is a common phenomenon during freeze-up seasons.

For oil-covered NI, we performed a backscatter sensitivity study to assess the impact of varying oil layer thickness on bare NI through backscatter simulations. The results in Fig. 3(a) and (b) revealed that the predicted signatures exhibited a highly sensitive response with decreasing NRCS values as the oil layer thickness increased. We attributed this to signal attenuation within the saline-oil layer due to a very large imaginary part of the saline-oil dielectrics. Past experiments by Firoozy et al. [40] and Asihene et al. [17] supported this finding. The study in [40] injected a larger volume of oil (20 L) beneath the ice, resulting in a thicker oil layer (value unknown) when the oil migrated onto the ice surface. Meanwhile the study in [17] injected a smaller oil volume (6 L) beneath the ice, leading to a thinner oil layer (5 mm) when the oil migrated onto the ice surface. Both studies reported a decrease in backscatter sensitivity, with the former demonstrating a more pronounced effect. Although our study is a simulation where oil is spilled on top of ice, the observations from these studies confirm the existence of complex physical processes at interplay between oil and sea ice, which contribute to the

	Output of inversion					
Adjusted variables (cm)	₫₁ (True=5 mm)	ε ₁ ('True'=26.53+j26.21)	ε ₂ ('True'=5.13+j1.62)	ε ₃ ('True'=5.82+j1.76)	Elapsed time (secs)	
d_2 (True= 0.5)						
0.1	5.97	45.18+j28.21	3.5+j1.19	5.86+j0.50	91	
0.3	6.55	38.65+j22.07	6+j2	5.35+j0.50	65	
0.7	4.94	52+j36.48	6+j0.50	4.26+j0.50	189	
1	5.14	51.81+j37.08	5.07+j0.52	3.54+j1.97	69	
d_3 (True= 4.5)						
1.5	6.34	32.70+j26.22	6+j0.50	5.99+j0.50	73	
2.5	5.43	22.37+j22.22	5.99+j0.51	3.50+j0.50	69	
8.5	5.10	52+j35.58	6+j2	3.55+j0.50	44	
9.5	5.22	51.78+j34.44	5.97+j1.92	3.64+j0.51	65	
$h_{12;} l_{12}$ ('True'= 0.05; 1)						
0.03*; 1	5.85	40.50+j22	6+j0.5	4.77+j0.50	115	
0.1*; 1	5.49	52+j42	6+j1.99	4.85+j0.50	103	
0.05; 0.5**	5.46	50.77+j28.22	6+j0.50	4.80+j0.50	128	
0.05; 2**	5.26	51.54+j34.21	6+j0.50	4.85+j0.50	117	
$0.1^{\dagger}; 2^{\dagger}$	9.14	52+j23.57	6+j0.50	4.86+j0.50	73	
0.03 [†] ; 0.5 [†]	2.80	22+j39.89	6+j0.50	4.86+j0.50	153	
$h_{3w;} l_{3w}$ ('True'= 0.2; 1)						
0.1*; 1	4.98	51.81+j35.72	5.97+j1.77	4.82+j0.51	77	
0.25*; 1	5.01	51.50+j36.53	5.99+j0.52	4.85+j0.50	57	
0.2; 0.8**	4.86	51.69+j37.36	5.99+j0.53	4.87+j0.50	70	
0.2; 2**	5.35	51.80+j33.45	6+j0.50	4.86+j0.50	134	
$0.25^{\dagger}; 2^{\dagger}$	5.31	51.32+j34.06	5.99+j0.52	4.85+j0.50	50	
$0.1^{\dagger}; 0.5^{\dagger}$	5.06	51.85+j34.91	5.97+j1.89	4.81+j0.51	72	

TABLE V Estimated Variables of the Oil-Covered NI Profile Using Erroneous Subsurface Ice Thickness and Roughness

All notations are defined in Table I. The bold highlight means that the primary estimated variable of interest is oil slick thickness ($d_1 = 5$ mm). The superscripts, * and ** represents independent adjustments made with either h or l, respectively. Meanwhile, the superscript, † represents adjustments made with both h and l.

smoothening of the NI surface and alteration of its dielectric properties.

For validation purposes, our simulation result compares well with the experimentally measured backscatter (at 24.5° incidence angle) of 5 mm oil-covered NI. This was expected as the presence of oil significantly reduces the roughness rmsheight, leading to a smooth oil/ice interface.

Furthermore, we hope to determine how much an increase in oil layer thickness suppresses the backscatter response closer to, and even below the NESZ level (-45 dB) of our C-band microwave scatterometer. This level was met at lower incidence angles ($\leq 35^{\circ}$) up to 5 mm oil thickness [see Fig. 3(b)]. Despite this, as shown in Fig. 3(a), the backscatter responses of the oil-covered ice (especially the 5-mm thick curve), were in close proximity to the NESZ background, leading to a poor SNR [25]. This indicates that a potential noise leakage from the receiver noise floor could compromise the usefulness of NRCS for oil spill detection [24]. Using the mild threshold (NESZ + 2 dB), preserve the acceptable signals at incidence angles $\leq 30^{\circ}$, while the harsh threshold (NESZ + 6 dB) preserves the useful signals at incidence angles $\leq 25^{\circ}$. These thresholds highlight that the NRCS signal values at low incidence angles are well-suitable for analyzing low backscatter phenomena (e.g., [44], [54]). Given our SNR analysis, we recommend the mild threshold for ground-based scatterometer, and the harsh threshold for airborne/spaceborne SAR measurements of oil-covered NI.

Comparing our NESZ level to that of radar satellites with high noise backgrounds (e.g., -33 to -38 dB for RCM ScanSAR low noise beam mode [45]), we speculate that our prediction could be limited to a slick layer of 2 mm or even lower. This highlights the need for technological advancements in the development of new SAR satellites with low NESZ levels, as well as the importance of conducting more in situ radar measurements for calibration purposes. Other factors we considered include the sensitivity of C-band radar over the range of incidence angles. As noted in Fig. 3(a) and (b), we found that the backscatter sensitivity is greater at lower incidence angles compared to higher incidence angles (>35°). This difference in sensitivity arises from the dominance of surface scattering at lower incidence angles, whereas specular reflection becomes more dominant at higher incidence angles.

To establish a relationship between the physical properties and backscatter sensitivity of variations of oil thickness on the dielectric profile of NI, we modeled the penetration depth of microwave propagation at C-band frequency (see Table III). Our findings indicate that in the absence of a saline-oil cover, microwave radiations can penetrate deeper into the oil-free NI, resulting in a higher backscatter response compared to oil-covered NI. This is primarily due to surface scattering contributions resulting from the strong dielectric contrast at the air/ice interface. On the other hand, when a saline oil is present, its surface weakens the dielectric mismatch and suppresses the surface scattering mechanism from the oil/ice interface. This phenomenon has been described by Asihene et al. [17] as a "masking effect" that leads to a constant backscatter response over time.

Based on our simulated microwave penetration depth data across each layer of the oil-covered NI (see Table III), we initially expected the backscatter sensitivity to approach or even fall below the NESZ level as the oil thickness increases above 2 mm. Surprisingly, we found that the backscatter sensitivity remained above the NESZ level until the surface oil thickness exceeded 5 mm [see Fig. 3(a) and (b)]. This indicates that even though the estimated penetration depth through the saline-oil layer is 2 mm (which is defined as the depth where the EM power drops by a factor of e = 2.7 times at zero degree incidence angle [22]), some fraction of EM energy still reaches the oil/ice rough surface at the 5 mm depth. As a result, this phenomenon generates a measurable backscatter signal higher than the NESZ level up to a certain incidence angle.

The sensitivity of the $\Upsilon_{VV/HH}$ to oil slick thickness variations provides a different perspective on our investigation because of its ability to discriminate between uncontaminated and oil-contaminated NI [43]. Prior to the oil spill, $\Upsilon_{VV/HH}$ exhibited a steep increase over the range of incidence angles. This behavior has been linked to highly saline brine cover under freeze-up conditions, which reflects more microwave energy at larger incidence angles, decreases horizontally polarized wave transmission, and increases the $\Upsilon_{VV/HH}$ values [55]. When oil is introduced, $\Upsilon_{VV/HH}$ decreases with a clear contrast from the oil-free ice, which we attributed to the reduction of the surface roughness effect. This finding agrees with the observation by Brekke et al. [54] which found that oil reduces the SPM-simulated $\Upsilon_{VV/HH}$ response.

In comparison to the NRCS, $\Upsilon_{VV/HH}$ displays inconsistent sensitivity to variations in oil thickness [refer to Fig. 3(a) and (b)]. While sensitivity becomes apparent upon oil introduction, it follows neither a decreasing nor increasing trend with further increments in oil thickness. This inconsistency could be attributed to the VV/HH ratio, which is less sensitive to the roughness effect compared to either the

VV or HH NRCS. This is the first time, to our knowledge, that such a result has been observed for oil-covered NI; thus, additional experiments data are required for validation. Moreover, when compared to NRCS, the $\Upsilon_{VV/HH}$ displayed a negligible contrast at lower incidence angles and very low contrast at higher incidence angles as the oil layer thickness increased [see Fig. 3(c)]. This indicate that the $\Upsilon_{VV/HH}$ data was insensitive to a changing oil slick thickness. As a result, we attest that inversion studies involving the retrieval of oil slick thickness on sea ice are more likely to be successful with remotely sensed NRCS observations.

B. Oil Thickness Retrieval

After interpreting the scattering sensitivity predicted by our multilayered-SPM forward solver, we focus our attention on the second question involving the inversion of oil layer thickness on bare, undeformed NI terrains. To tackle this, we used two different backscatter datasets during the PSO-based parametric inversion process to evaluate the data misfit objective function.

Initially, we employed synthetic inputs to test the applicability of our optimizer. The results in Table IV demonstrate that when using NRCS, the estimated oil thickness was closer to the true value compared to when using $\Upsilon_{VV/HH}$. This indicates that the PSO hyperparameters were optimally finetuned, and NRCS datasets are suitable for accurately retrieving oil thickness, as their predicted signatures had shown to be sensitive to the increasing oil layer thickness on ice [see Fig. 4(a)]. Our findings agree with several previous inversion studies that utilized NRCS datasets for their inversion studies (e.g., [12], [31], [56] and references therein). On the contrary, the inversion using $\Upsilon_{VV/HH}$ was unsuccessful because the simulation results were inconsistent and lacked contrast with oil thickness variations. We discussed in the preceding Section IV-A that this inconsistency and lack of contrast was caused by the derivation of $\Upsilon_{VV/HH}$, which is less sensitive to roughness, making it solely dependent on CDCs of the inhomogeneous medium and radar incidence angle [23]. This highlights the significance of roughness parameters for accurate estimation of oil thickness on sea ice; thus, we recommend using instruments such as LiDAR for precise measurement of surface roughness [57], as well as roughness at the underlying oil/ice and ice/water interfaces (although the technology is currently unavailable).

In addition, we used experimentally measured NRCS data to validate the performance of PSO, and the estimated result closely matched the true oil thickness (see Table IV). Compared to synthetic data that employed three incidence angles, the estimation error difference result was impressive with the measured NRCS data at a single incidence angle (see Fig. 4). We acknowledged that using a single dataset in terms of incidence angle and oil thickness is insufficient because our studies were limited to experimental designs constraints. Our future work will aim to improve the realism of the inverse solver by expanding the inversion method to encompass a wide range of incidence angles and oil layer thicknesses.

Our PSO-based parametric inversion technique allows for quantitative extraction of multiple physical parameters from remote sensing observations. Although oil slick thickness is our primary parameter of interest, we were able to retrieve the CDCs of saline oil, oiled frazil, and sea ice (see Table IV). These retrievable CDCs were randomly close to their true values within a reasonable margin of error (except for the 5 mm oil thickness estimation, which we attributed to the noise leakage from the NESZ level). We say that the error margin is "reasonable" because the environmental conditions that contribute to sea ice formation in regions including the Arctic, undergo dynamic processes [58], resulting in constant changes in the thermophysical properties that characterize these inverted CDCs. As a result, we anticipated these random variations, but we recommend that evaluations of the objective function be monitored. For instance, as shown in Fig. 4(b), the convergence of the objective function demonstrates a guaranteed optimal value of our estimated parameter of interest. Moreover, we recommend that future work consider direct dielectric measurements of saline-oil and oil-contaminated ice in a wide range of temperatures and salinities. These measurements could be used for validating the inversion algorithms and also for tuning/validating the dielectric mixture model.

In time-sensitive emergency situations such as oil spills, rapid identification of the thickest oil spill zone is crucial for mounting an effective response. This necessitates the use of fast inverse scattering models capable of extracting the oil thickness from remote sensing observations. Our inversion method takes an iterative approach that uses the forward solver to assess the data misfit objective function, while relying on the PSO algorithm to minimize error differences in the estimated unknown variables. As shown in Table IV, the computational cost of our method is extremely low, taking less than 5 min to complete (on a standard PC with 8 GB RAM and 1.6 GHz 4-core CPUs). This run time demonstrates the efficiency of our forward solver and its potential to handle larger radar datasets (such as SAR imagery).

Our inversion studies have been successful thus far; however, it is important to recognize the limitations of this method. First, this inversion process does not support a zerooil thickness. We assumed that oil must be present, which implies that the occurrence of an oil spill situation should be detected before its application. Second, this inversion method is only generalized for oil covering bare NI and does not account for the presence of frost flower and snow cover, which are commonly associated with NI growth [20]. This limitation arises from insufficient snowfall and unfavorable atmospheric conditions for frost flower formation during the SERF oilon-ice experiment [17]. Our future work will consider this limitation and introduce surface cover such as frost flower, snow, or combination of both into the multilayer dielectric profile NI. Third, this inversion method relies on prior knowledge associated with the inhomogeneous multilayered oil-covered NI (e.g., roughness, ice thickness, and water dielectrics) to improve estimation accuracy. In practice, the existing radar satellites such as SAR, lack the capability to directly retrieve (at least with the limited number of measurements) some of these input parameters, including the roughness at the oil/ice and ice/water interfaces, as well as the underlying ice thicknesses of the oil-covered NI (see Fig. 2). This implies that there is a likelihood of using inaccurate a priori information, which could lead to estimation errors.

For this reason, we tested the estimation accuracy of our inversion with erroneous a priori subsurface roughness and ice thickness (see Table V); note that no testing was done with incorrect surface roughness because oil spills on NI scenes usually smoothens the ice surface [17]. Our findings emphasize the importance of using accurate a priori knowledge for precise oil thickness retrievals in oil-covered NI profiles. Understanding these relationships, we establish that PSO-inversion method achieves optimal results when NI thickness is between 2 and 10 cm, with the topmost layer (frazil) thickness falling between 0.1 and 1 cm. Furthermore, we found that the roughness at the oil/ice interface can significantly affect the inversion process compared to the ice/water interface. This is because the presence of highly saline-oil cover attenuates high microwave signals, limiting the microwave propagation through the sea ice medium (see Table III).

VI. CONCLUSION

In this manuscript, we presented an EM modeling strategy for retrieving the thickness of an experimental oil spill on sea ice. Our approach involves using a first-order approximation of a multilayered-SPM forward solver to simulate the C-band radar backscatter from an oil-covered NI and employing a PSO-based parametric inverse solver to estimate the oil slick thickness.

Prior to our simulation and inversion analyses, we checked the influence of noise contamination using mild (2 dB) and harsh (6 dB) thresholds of the measured and simulated NRCS data above the NESZ level (see Fig. 3). Our results showed that NRCS data for the 1, 3, and 5 mm slick layer curves at lower incidence angles $\leq 25^{\circ}$ were reliable for analysis. This finding highlights the need for technological advancements in the development of new SAR satellites with low NESZ levels.

Our simulation results demonstrate several important findings (see Fig. 3). First, the predicted backscatter is highly sensitive to the increase in oil layer thickness, resulting in a continuous reduction in backscatter strength. This reduction is attributed to signal attenuation within the saline-oil layer. Second, the copolarization ratio is initially sensitive as soon as oil covers the NI surface, decreasing its magnitudes. However, unlike the backscatter sensitivity, the copolarization ratio lacks consistency due to the reduction of the roughness effect. Third, the penetration depth through each layer of the oil-covered NI is significantly hindered by the oil layer, which contains highly saline brine wicking from the topmost layer of NI.

Regarding our inversion results, the estimated oil thickness closely matches the true value (see Table IV). For instance, the synthetic NRCS underestimates 1 and 3 mm oil thicknesses by 3% and 8%, and overestimates 5 mm thickness by 25%, while the experimental NRCS overestimates 5 mm oil thickness by 8%. The corresponding retrievable dielectrics exhibit random variations within a reasonable margin of error, attributed to the dynamic nature of NI thermophysical properties. With the use of synthetic copolarization ratio, the estimated oil thickness proves unsuccessful, and we recommend against its use.

These findings hold significant implications for oil spill response efforts. Accurately estimating the thickness of an oil slick allows for the identification of areas with the highest concentration of oil, which should be prioritized for immediate action using remediation techniques such as in situ burning and chemical dispersants. By swiftly providing this information, the disaster response team can offer actionable methods and recommendations for containing and cleaning up the spill. In the context of oil spills, time is of the essence, and the rapid identification of the thickest oil spill zones is crucial for making well-informed decisions that minimize damage to natural resources, communities, and economies dependent on them. Additionally, combining these thickness estimates with the extent of the spill area enables more precise calculations of the total volume of oil spilled, facilitating accurate assessments of environmental impacts.

While our study establishes the foundation for using C-band radar observation to estimate oil slick thickness on NI, there is still room for improvement. First, our inversion method relied on a single experimental data set for validation. Second, the method assumes the presence of oil, making it applicable primarily as a planning tool for oil spill situations. Third, the a priori parameters used in our inversion process cannot be directly extracted from current radar satellites. However, this is not a problem provided our assumptions for subsurface NI thickness are within 10 cm, and for underlying roughness remain within the SPM region of validity (see Table V). Finally, surface features such as frost flower and snow, which commonly occur during sea ice formation, were not considered in our inversion method. In future research, we intend to address these limitations by incorporating a more generalized scenario encompassing various surface features and a wider range of experimental data.

ACKNOWLEDGMENT

The authors appreciate the support of Dave Binne (SERF Technician, CEOS), M. C. Fuller, M. L. Harasyn, D. Landry, D. S. Desmond, and K. Polcwiartek.

REFERENCES

- B. Ellis and L. Brigham. (2009). Arctic Marine Shipping Assessment Report. Accessed: Dec. 30, 2021. [Online]. Available: http://hdl. handle.net/11374/54
- [2] J. Dawson, L. Pizzolato, S. E. L. Howell, L. Copland, and M. E. Johnston, "Temporal and spatial patterns of ship traffic in the Canadian Arctic from 1990 to 2015," *Arctic*, vol. 71, no. 1, pp. 15–26, 2018, doi: 10.14430/arctic4698.
- [3] A. K. Y. Ng, J. Andrews, D. Babb, Y. Lin, and A. Becker, "Implications of climate change for shipping: Opening the Arctic Seas," WIREs Climate Change, vol. 9, no. 2, pp. 1–18, Mar. 2018, doi: 10.1002/wcc.507.
- [4] B. M. Boylan and D. Elsberry. (2019). Climate Change and Maritime Traffic in the Arctic. The Center for Arctic Policy Studies. Accessed: Oct. 2, 2021. [Online]. Available: https://uaf.edu/caps/our-work/policy-perspectives-files/climate-changeand-maritime-traffic-in-the-arctic-pb4-final-12dec2019.pdf
- [5] D. G. Barber, "Microwave remote sensing, sea ice and Arctic climate," *Phys. Canada*, vol. 61, no. 5, pp. 105–111, 2005.
- [6] C. Brekke and A. H. S. Solberg, "Oil spill detection by satellite remote sensing," *Remote Sens. Environ.*, vol. 95, no. 1, pp. 1–13, Mar. 2005, doi: 10.1016/j.rse.2004.11.015.
- [7] K. C. Jezek et al., "A broad spectral, interdisciplinary investigation of the electromagnetic properties of sea ice," *IEEE Trans. Geosci. Remote Sens.*, vol. 36, no. 5, pp. 1633–1641, Sep. 1998, doi: 10.1109/36.718635.

- [8] W. Weeks and W. D. Hibler, On Sea Ice. Fairbanks, AK, USA: Univ. Alaska Press, 2010.
- [9] D. Isleifson, M. L. Harasyn, D. Landry, D. Babb, and E. Asihene, "Observations of thin first year sea ice using a suite of surface radar, LiDAR, and drone sensors," *Can. J. Remote Sens.*, vol. 49, no. 1, pp. 1–24, Jul. 2023, doi: 10.1080/07038992.2023.2226220.
- [10] W. F. Weeks and S. F. Ackley, "The growth, structure, and properties of sea ice," in *The Geophysics of Sea Ice*. Boston, MA, USA: Springer, 1986, pp. 9–164.
- [11] D. K. Perovich and J. A. Richter-Menge, "Surface characteristics of lead ice," J. Geophys. Res., Oceans, vol. 99, no. C8, pp. 16341–16350, Aug. 1994, doi: 10.1029/94jc01194.
- [12] S. V. Nghiem, R. Kwok, S. H. Yueh, and M. R. Drinkwater, "Polarimetric signatures of sea ice: 2. Experimental observations," *J. Geophys. Res., Oceans*, vol. 100, no. C7, pp. 13681–13698, Jul. 1995, doi: 10.1029/95jc00938.
- [13] S. Potter, I. Buist, K. Trudel, D. Dickins, and E. Owens. (2012). Spill Response in the Arctic Offshore. The American Petroleum Institute. Accessed: Oct. 10, 2022. [Online]. Available: https://www.api.org//media/Files/EHS/Clean_Water/Oil_Spill_Prevention/Spill-Response-inthe-Arctic-Offshore.pdf
- [14] P. Gloersen and J. K. Larabee, "An optical model for the microwave properties of sea ice," NASA Goddard Space Flight Center, Greenbelt, MD, USA, Tech. Rep. 83865, Nov. 1981. [Online]. Available: https:// ntrs.nasa.gov/api/citations/19820009687/downloads/19820009687.pdf
- [15] D. Isleifson, A. S. Komarov, D. Desmond, G. Stern, and D. Barber, "Modeling backscatter from oil-contaminated sea ice using a multi-layered scattering model," in *Proc. IEEE Int. Geosci. Remote Sens. Symp.*, Sep. 2020, pp. 3031–3034, doi: 10.1109/IGARSS39084.2020.9323887.
- [16] A. Mansoori, D. Desmond, G. Stern, and D. Isleifson, "Modeling normalized radar cross-section of oil-contaminated sea ice with small perturbation method," in *Proc. IEEE 19th Int. Symp. Antenna Technol. Appl. Electromagn. (ANTEM)*, Aug. 2021, pp. 1–2, doi: 10.1109/ANTEM51107.2021.9519056.
- [17] E. Asihene et al., "Toward the detection of oil spills in newly formed sea ice using C-band multipolarization radar," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 4302615, doi: 10.1109/TGRS.2021.3123908.
- [18] E. Asihene, A. S. Komarov, G. A. Stern, C. Gilmore, and D. Isleifson, "Simulating C-band radar backscatter of oil-covered new ice in a Mesocosm," in *Proc. URSI Int. Symp. Electromagn. Theory*, 2023, pp. 1–3.
- [19] A. Sihvola, *Electromagnetic Mixing Formulas and Applications*. London, U.K.: Institution of Engineering and Technology, 1999.
- [20] R. G. Onstott, "SAR and scatterometer signatures of sea ice," in *Microwave Remote Sensing of Sea Ice*, F. D. Carsey, Ed. Washington, DC, USA: Amer. Geophys. Union, 1992, pp. 73–104.
- [21] M. Hallikainen and D. P. Winebrenner, "The physical basis for sea ice remote sensing," in *Microwave Remote Sensing of Sea Ice*, F. D. Carsey, Ed. Washington, DC, USA: Amer. Geophys. Union, 1992, pp. 29–46.
- [22] F. Ulaby and D. Long, *Microwave Radar and Radiometric Remote Sensing*. Ann Arbor, MI, USA: The Univ. Michigan Press, 2014.
- [23] D. P. Winebrenner, L. D. Farmer, and I. R. Joughin, "On the response of polarimetric synthetic aperture radar signatures at 24-cm wavelength to sea ice thickness in Arctic leads," *Radio Sci.*, vol. 30, no. 2, pp. 373–402, Mar. 1995, doi: 10.1029/94RS02313.
- [24] B. Minchew, C. E. Jones, and B. Holt, "Polarimetric analysis of backscatter from the deepwater horizon oil spill using L-band synthetic aperture radar," *IEEE Trans. Geosci. Remote Sens.*, vol. 50, no. 10, pp. 3812–3830, Oct. 2012, doi: 10.1109/TGRS.2012.2185804.
- [25] S. Angelliaume et al., "SAR imagery for detecting sea surface slicks: Performance assessment of polarization-dependent parameters," *IEEE Trans. Geosci. Remote Sens.*, vol. 56, no. 8, pp. 4237–4257, Aug. 2018, doi: 10.1109/TGRS.2018.2803216.
- [26] S. Skrunes, C. Brekke, C. E. Jones, M. M. Espeseth, and B. Holt, "Effect of wind direction and incidence angle on polarimetric SAR observations of slicked and unslicked sea surfaces," *Remote Sens. Environ.*, vol. 213, pp. 73–91, Aug. 2018, doi: 10.1016/j.rse.2018.05.001.
- [27] M. M. Espeseth, C. Brekke, C. E. Jones, B. Holt, and A. Freeman, "The impact of system noise in polarimetric SAR imagery on oil spill observations," *IEEE Trans. Geosci. Remote Sens.*, vol. 58, no. 6, pp. 4194–4214, Jun. 2020, doi: 10.1109/TGRS.2019.2961684.
- [28] A. M. Johansson, M. M. Espeseth, C. Brekke, and B. Holt, "Can mineral oil slicks be distinguished from newly formed sea ice using synthetic aperture radar?" *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 13, pp. 4996–5010, 2020, doi: 10.1109/JSTARS.2020.3017278.

- [29] D. P. Winebrenner et al., "Microwave sea ice signature modeling," in *Microwave Remote Sensing of Sea Ice*, F. D. Carsey, Ed. Washington, DC, USA: Amer. Geophys. Union, 1992, pp. 137–175.
- [30] D. G. Barber and E. F. LeDrew, "On the links between microwave and solar wavelength interactions with snow-covered first-year sea ice," *Arctic*, vol. 47, no. 3, pp. 298–309, Jan. 1994, doi: 10.14430/arctic1300.
- [31] R. Kwok, S. Nghiem, S. H. Yueh, and D. D. Huynh, "Retrieval of thin ice thickness from multifrequency polarimetric SAR data," *Remote Sens. Environ.*, vol. 51, no. 3, pp. 361–374, Mar. 1995, doi: 10.1016/0034-4257(94)00017-H.
- [32] E. M. Nassar, J. T. Johnson, and R. Lee, "A numerical model for electromagnetic scattering from sea ice," *IEEE Trans. Geosci. Remote Sens.*, vol. 38, no. 3, pp. 1309–1319, May 2000, doi: 10.1109/36.843024.
- [33] K. M. Golden et al., "Forward electromagnetic scattering models for sea ice," *IEEE Trans. Geosci. Remote Sens.*, vol. 36, no. 5, pp. 1655–1674, Apr. 1998, doi: 10.1109/36.718637.
- [34] D. Isleifson, I. Jeffrey, L. Shafai, J. LoVetri, and D. G. Barber, "A Monte Carlo method for simulating scattering from sea ice using FVTD," *IEEE Trans. Geosci. Remote Sens.*, vol. 50, no. 7, pp. 2658–2668, Jul. 2012, doi: 10.1109/TGRS.2011.2173940.
- [35] A. K. Fung and K. S. Chen, *Microwave Scattering and Emission Models for Users*. Norwood, MA, USA: Artech House, 2010.
- [36] A. S. Komarov, L. Shafai, and D. G. Barber, "Electromagnetic wave scattering from rough boundaries interfacing inhomogeneous media and application to snow-covered sea ice," *Prog. Electromagn. Res.*, vol. 144, pp. 201–219, 2014, doi: 10.2528/PIER13111209.
- [37] A. S. Komarov, D. Isleifson, D. G. Barber, and L. Shafai, "Modeling and measurement of C-band radar backscatter from snow-covered firstyear sea ice," *IEEE Trans. Geosci. Remote Sens.*, vol. 53, no. 7, pp. 4063–4078, Jul. 2015, doi: 10.1109/TGRS.2015.2390192.
- [38] A. S. Komarov, J. C. Landy, S. A. Komarov, and D. G. Barber, "Evaluating scattering contributions to C-band radar backscatter from snow-covered first-year sea ice at the winter–spring transition through measurement and modeling," *IEEE Trans. Geosci. Remote Sens.*, vol. 55, no. 10, pp. 5702–5718, Oct. 2017, doi: 10.1109/TGRS.2017.2712519.
- [39] M. Oristaglio and H. Blok, "Wavefield imaging and inversion in electromagnetics and acoustics," Centre Tech. Geosci., Delft Univ. Technol., Delft, The Netherlands, Tech. Rep. Et/EM 1995-21, 1995, p. 167.
- [40] N. Firoozy et al., "An electromagnetic detection case study on crude oil injection in a young sea ice environment," *IEEE Trans. Geosci. Remote Sens.*, vol. 55, no. 8, pp. 4465–4475, Aug. 2017, doi: 10.1109/TGRS.2017.2692734.
- [41] N. Firoozy et al., "A controlled experiment on oil release beneath thin sea ice and its electromagnetic detection," *IEEE Trans. Geosci. Remote Sens.*, vol. 56, no. 8, pp. 4406–4419, Aug. 2018, doi: 10.1109/TGRS.2018.2818717.
- [42] E. Asihene et al., "C-band backscatter of oil-polluted new sea ice in a Mesocosm," in Proc. IEEE 19th Int. Symp. Antenna Technol. Appl. Electromagn. (ANTEM), Aug. 2021, pp. 1–2, doi: 10.1109/ANTEM51107.2021.9518483.
- [43] E. Asihene, G. Stern, D. G. Barber, C. Gilmore, and D. Isleifson, "Toward the discrimination of oil spills in newly formed sea ice using C-band radar polarimetric parameters," *IEEE Trans. Geosci. Remote Sens.*, vol. 61, 2023, Art. no. 4300115, doi: 10.1109/TGRS.2022.3232083.
- [44] C. Quigley, C. Brekke, and T. Eltoft, "Comparison between dielectric inversion results from synthetic aperture radar co- and quadpolarimetric data via a polarimetric two-scale model," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 4200218, doi: 10.1109/TGRS.2020.3038366.
- [45] A. S. Komarov, S. A. Komarov, and M. Buehner, "Ocean surface wind speed retrieval from the RADARSAT constellation mission coand cross-polarization images without wind direction input," *IEEE Trans. Geosci. Remote Sens.*, vol. 60, 2022, Art. no. 4205415, doi: 10.1109/TGRS.2021.3137127.
- [46] D. Isleifson et al., "A multi-frequency suite of polarimetric scatterometers for Arctic remote sensing," in *Proc. URSI Int. Symp. Electromagn. Theory*, 2023, pp. 1–3.
- [47] G. F. N. Cox and W. F. Weeks, "Equations for determining the gas and brine volumes in sea-ice samples," J. Glaciol., vol. 29, no. 102, pp. 306–316, 1983.
- [48] B. Scheuchl, R. Caves, D. Flett, R. De Abreu, M. Arkett, and I. Cumming, "The potential of cross-polarization information for operational sea ice monitoring," in *Proc. Envisat & ERS Symp.*, 2005, pp. 2139–2145.

- [49] M. Shokr and M. Dabboor, "Polarimetric SAR applications of sea ice: A review," *IEEE J. Sel. Topics Appl. Earth Observ. Remote Sens.*, vol. 16, pp. 6627–6641, 2023, doi: 10.1109/JSTARS.2023.3295735.
- [50] J. Kennedy and R. Eberhart, "Particle swarm optimization," in Proc. Int. Conf. Neural Netw., vol. 4, 1995, pp. 1942–1948, doi: 10.1109/ICNN.1995.488968.
- [51] S. Ebbesen, P. Kiwitz, and L. Guzzella, "A generic particle swarm optimization MATLAB function," in *Proc. Amer. Control Conf. (ACC)*, Jun. 2012, pp. 1519–1524, doi: 10.1109/ACC.2012.6314697.
- [52] A. Mansoori, D. Isleifson, D. Desmond, and G. Stern, "Development of dielectric measurement techniques for Arctic oil spill studies," in *Proc. IEEE Int. Symp. Antennas Propag. North Amer. Radio Sci. Meeting*, Jul. 2020, pp. 883–884, doi: 10.1109/IEEECONF35879.2020.9329458.
- [53] S. A. Komarov, A. S. Komarov, D. G. Barber, M. J. L. Lemes, and S. Rysgaard, "Open-ended coaxial probe technique for dielectric spectroscopy of artificially grown sea ice," *IEEE Trans. Geosci. Remote Sens.*, vol. 54, no. 8, pp. 4941–4951, Aug. 2016, doi: 10.1109/TGRS.2016.2553110.
- [54] C. Brekke, B. Holt, C. Jones, and S. Skrunes, "Discrimination of oil spills from newly formed sea ice by synthetic aperture radar," *Remote Sens. Environ.*, vol. 145, pp. 1–14, Apr. 2014, doi: 10.1016/j.rse.2014.01.015.
- [55] S. V. Nghiem et al., "Evolution in polarimetric signatures of thin saline ice under constant growth," *Radio Sci.*, vol. 32, no. 1, pp. 127–151, Jan. 1997, doi: 10.1029/96RS03051.
- [56] Y. J. Lee, K. C. Yeong, and H.-T. Ewe, "A study of an inversion model for sea ice thickness retrieval using simulated annealing," *Prog. Electromagn. Res. C*, vol. 104, pp. 143–155, 2020, doi: 10.2528/PIERC20042001.
- [57] J. C. Landy, D. Isleifson, A. S. Komarov, and D. G. Barber, "Parameterization of centimeter-scale sea ice surface roughness using terrestrial LiDAR," *IEEE Trans. Geosci. Remote Sens.*, vol. 53, no. 3, pp. 1271–1286, Mar. 2015, doi: 10.1109/TGRS.2014.2336833.
- [58] W. B. Tucker III, D. K. Perovich, A. J. Gow, W. F. Weeks, and M. R. Drinkwater, "Physical properties of sea ice relevant to remote sensing," in *Microwave Remote Sensing of Sea Ice*, F. D. Carsey, Ed. Washington, DC, USA: Amer. Geophys. Union, 1992 pp. 9–28.



Elvis Asihene (Graduate Student Member, IEEE) received the B.Sc. degree in geological engineering from the Ghana's University of Mines and Technology, Tarkwa, Ghana, in 2011, and the M.Sc. degree in applied geophysics from the University of Ghana, Accra, Ghana, in 2017. He is currently pursuing the Ph.D. degree in geography with the University of Manitoba, Winnipeg, MB, Canada.

His main areas of research are in arctic science, remote sensing, and GIS. He is particularly interested in using microwave remote sensing to detect

oil spills in Arctic ice-covered waters. His other research interests include electromagnetic modeling and inversion of sea-ice-related parameters, as well as the application of polarimetry and interferometry radar for target characterization and surface subsidence, respectively.



Alexander S. Komarov (Senior Member, IEEE) received the B.Sc. degree (Hons.) in radiophysics and electronics and the M.Sc. degree (Hons.) in physics from Altai State University, Barnaul, Russia, in 2006 and 2008, respectively, and the Ph.D. degree in electrical engineering from the University of Manitoba, Winnipeg, MB, Canada, in 2015.

Since 2015, he has been a Research Scientist with the Meteorological Research Division, Environment and Climate Change Canada, Ottawa, ON, Canada. His research interests include synthetic aperture

radar (SAR) remote sensing of Arctic Sea ice and the ocean surface and SAR data assimilation in numerical systems with particular focus on the RADARSAT Constellation Mission.



Gary Stern received the Ph.D. degree in chemistry on mass spectrometry from the University of Manitoba, Winnipeg, MB, Canada, in 1991.

He is currently a Professor with the Centre for Earth Observations Sciences, University of Manitoba. He was involved in the study of environmental pathways of contaminants in Arctic marine and freshwater ecosystems. He is involved in the development of oil detection and remediation technologies, impacts, and fate of oil spills in Arctic marine environments. He is currently the co-lead

of the GENICE II project that involves genomics research in oil spill preparedness and emergency response in an Arctic marine environment.



Dustin Isleifson (Senior Member, IEEE) received the Ph.D. degree in electrical and computer engineering from the University of Manitoba, Winnipeg, MB, Canada, in 2011.

From 2013 to 2016, he was an Electrical Engineer with Magellan Aerospace, Winnipeg, where he worked on the Radarsat Constellation Mission (RCM). He has been with the Department of Electrical and Computer Engineering and the Centre for Earth Observation Science (CEOS), University of Manitoba, as an Assistant Professor in 2016 and as

an Associate Professor in 2022. His research interests remote sensing, Arctic science, antenna design, and satellite technologies.

Dr. Isleifson is the Chapter Chair of the Winnipeg Section of the IEEE GRS/AES.



Colin Gilmore (Senior Member, IEEE) is currently the Tier 2 Canada Research Chair in Applied Electromagnetic Inversion and an Assistant Professor of electrical and computer engineering with the University of Manitoba, Winnipeg, MB, Canada.

His research interests involve the systems and algorithms of sensing objects with electromagnetic waves. This includes biomedical imaging for cancer detection, geophysical images, and imaging of the contents of grain storage bins.